Author's response to reviews

Title: GDNF protects apoptosis in enteric glia: Evidence for an autocrine loop

Authors:

Martin Steinkamp (steinkam@med.uni-marburg.de)
Heike Gundel (heike.gundel@uniklinik-ulm.de)
Nadine Schulte (nadine.schulte@uniklinik-ulm.de)
Ulrike Spaniol (ulrike.spaniol@uniklinik-ulm.de)
carolin pflueger (carolin.pflueger@uniklinik-ulm.de)
Georg BT von Boyen (georg.boyen@uniklinik-ulm.de)

Version: 2 Date: 16 May 2011

Author's response to reviews:

In the revised version the Ethics/consent, Competing interest, Authors contributions and Acknowledgements are included